Author’s response to reviews

Title: Bilateral acute angle-closure glaucoma following Tramadol subcutaneous administration

Authors:

Anis Mahmoud (mahmoudanisss@hotmail.fr)
Fatma Abid (fatmabid@gmail.com)
Imen Ksiaa (khay.imen@yahoo.fr)
Sourour Zina (sourourzina@hotmail.com)
Riadh Messaoud (messaoud_riadh@yahoo.fr)
Moncef Khairallah (moncef.khairallah@yahoo.fr)

Version: 2 Date: 28 Jan 2018

Author’s response to reviews:

Dear Editors, Dear Reviewers,

Thank you for allowing us to revise the manuscript BOPH-D-17-00537 entitled “Bilateral acute angle closure glaucoma following tramadol subcutaneous administration” we submitted for publication in BMC Ophthalmology.

We thank the reviewers for their constructive and useful comments. Improvement of the language usage was made. Authors’ contribution section was edited. We made amendments to the manuscript accordingly.

Sincerely yours,

Moncef Khairallah, MD